Author's response to reviews

Title: Integrating intensified case finding of tuberculosis into HIV care: an evaluation from rural Swaziland

Authors:

Susan Elden (susan_elden@yahoo.co.uk)
Timothy Lawes (timlawesinindia@yahoo.co.uk)
Soren Kudsk-Iversen (sorenki@googlemail.com)
Joris Vandelanotte (jv2205@columbia.edu)
Sabelo Nkawanyana (sabeloswo@gmail.com)
William Welfare (wwelfare@nhs.net)
John Walley (j.walley@leeds.ac.uk)
John Wright (john.wright@bradfordhospitals.nhs.uk)

Version: 5 Date: 21 May 2011

Author's response to reviews:

Dear Editors,

I accept your request for formatting changes and have revised each of the areas you have highlighted.

Thank you,

Susan Elden